

Microbiology Spectrum

Recombinase-Aided Amplification Assay for Rapid Detection of Hypervirulent *Klebsiella pneumoniae* (hvKp) and Characterization of the hvKp Pathotype

Chao Yan, Yao Zhou, Shuheng Du, Bing Du, Hanqing Zhao, Yanling Feng, Guanhua Xue, Jinghua Cui, Lin Gan, Junxia Feng, Zheng Fan, Tongtong Fu, Ziying Xu, Qun Zhang, Rui Zhang, Xiaohu Cui, Ziyan Tian, Yujie Chen, Ting Zhang, Lei Huang, and Jing Yuan

Corresponding Author(s): Jing Yuan, Capital Institute of Pediatrics, and Lei Huang, Fifth Medical Center of Chinese PLA General Hospital

Review Timeline:

Submission Date: Editorial Decision: Revision Received: Accepted: October 4, 2022 January 9, 2023 February 6, 2023

February 17, 2023

Editor: MARK PANDORI

Reviewer(s): Disclosure of reviewer identity is with reference to reviewer comments included in decision letter(s). The following individuals involved in review of your submission have agreed to reveal their identity: Peng Zhu (Reviewer #2)

Transaction Report:

(Note: With the exception of the correction of typographical or spelling errors that could be a source of ambiguity, letters and reports are not edited. The original formatting of letters and referee reports may not be reflected in this compilation.)

DOI: https://doi.org/10.1128/spectrum.03984-22

1st Editorial Decision January 9,

20231

January 9, 2023

Prof. Jing Yuan
Capital Institute of Pediatrics
Department of Bacteriology
No. 2 yabao road, Chaoyang District
Beijing, Beijing 100020
China

Re: Spectrum03984-22 (Recombinase-Aided Amplification Assay for Rapid Detection of Hypervirulent Klebsiella pneumoniae (hvKp))

Dear Prof. Jing Yuan:

Thank you for submitting your manuscript to Microbiology Spectrum. When submitting the revised version of your paper, please provide (1) point-by-point responses to the issues raised by the reviewers as file type "Response to Reviewers," not in your cover letter, and (2) a PDF file that indicates the changes from the original submission (by highlighting or underlining the changes) as file type "Marked Up Manuscript - For Review Only". Please use this link to submit your revised manuscript - we strongly recommend that you submit your paper within the next 60 days or reach out to me. Detailed instructions on submitting your revised paper are below.

Link Not Available

Below you will find instructions from the Microbiology Spectrum editorial office and comments generated during the review.

ASM policy requires that data be available to the public upon online posting of the article, so please verify all links to sequence records, if present, and make sure that each number retrieves the full record of the data. If a new accession number is not linked or a link is broken, provide production staff with the correct URL for the record. If the accession numbers for new data are not publicly accessible before the expected online posting of the article, publication of your article may be delayed; please contact the ASM production staff immediately with the expected release date.

The ASM Journals program strives for constant improvement in our submission and publication process. Please tell us how we can improve your experience by taking this quick <u>Author Survey</u>.

Sincerely,

MARK PANDORI

Editor, Microbiology Spectrum

Journals Department American Society for Microbiology 1752 N St., NW Washington, DC 20036 E-mail: spectrum@asmusa.org

While we are willing to consider a revised version of this paper at Spectrum, it would be in your best interest to improve the writing. I recommend that you ask a colleague of yours who is a native English speaker to read and provide you some feedback on the writing. You are also welcome to use one of the services here: https://journals.asm.org/content/language-editing-services

Reviewer comments:

Reviewer #1 (Comments for the Author):

Manuscript is sound in methodology and analysis. Appropriate controls were implemented.

Line 74 Missing space between "siderophore" and "for"

Line 133 lists RAA sensitivity as 20 copies while line 239 lists 10 copies

Reviewer #2 (Comments for the Author):

The authors developed a new method to rapidly detect hvKp in clinical samples based on recombinase-aided amplification, and thought this approach is a powerful tool to solve the challenge of timely detection of hvKp infections in clinical and surveillance settings. There are comments which the authors should clarify.

Maior comments:

- 1. Considerable English revision is needed.
- 2. The title is focused on the rapid detection of hvKp. However, in manuscript, the authors also have written much information about phenotype and molecular characteristics of hvKp. These contents are not in accordance with the topic of this manuscript. Please clarify this.
- 3. There is on hvKp/cKp standard strains were used to evaluate the specificity in the study. It is not enough to use only plasmid containing genes peg-344 and rmpA, at least two standard strains of hvKp and cKp should also be examined.
- 4. It is not convincing to evaluate the RAA assay in clinical detection of hvKp only using clinical isolates. I suggest the authors to check the ability of the developed assay from clinical samples, such as, blood, saliva and nasopharyngeal swabs.
- 5. The authors should analyze the economics of this assay and explain in discussion, since the authors thought the cost of RAA assay is lower than real-time PCR or other nucleic acid-based detection methods in discussion.

 Minor comments:
- 1. Line 108: Delete the "peg-344 and rmpA"
- 2. Line 114: Delete the "using the RAA assay developed"
- 3. Line 114-116: Refine and rewrite this sentence.
- 4. Line 117: Change "The optimal reaction temperature range" to "The range of the reaction temperature".
- 5. Line 121: Change "Fluorescence" to "Fluorescent".
- 6. Line 123: Delete "evaluated"
- 7. Line 126: Add "analysis" before the word "of"
- 8. Line 130: The results from conventional PCR and real-time PCR could not be reviewed in Figure 3. Suggest the authors add these figures in Figure 3.
- 9. Line 131: Here, the unit of the concentration of recombinant plasmid is "copies/reaction", which is not in line with "copies/µl" in Line 135. Please the authors clarify.
- 10. Line 141-143: Refine and rewrite this sentence.
- 11. Line 148: Delete the "assay" before the word "results".
- 12. Line 149: Delete "which".
- 13. Line 159-161: Refine and rewrite this sentence.
- 14. Line 198-199: Refine and rewrite this sentence.
- 15. Line 202-204: Refine and rewrite this sentence.

Staff Comments:

Preparing Revision Guidelines

To submit your modified manuscript, log onto the eJP submission site at https://spectrum.msubmit.net/cgi-bin/main.plex. Go to Author Tasks and click the appropriate manuscript title to begin the revision process. The information that you entered when you first submitted the paper will be displayed. Please update the information as necessary. Here are a few examples of required updates that authors must address:

- Point-by-point responses to the issues raised by the reviewers in a file named "Response to Reviewers," NOT IN YOUR COVER LETTER.
- Upload a compare copy of the manuscript (without figures) as a "Marked-Up Manuscript" file.
- Each figure must be uploaded as a separate file, and any multipanel figures must be assembled into one file.
- Manuscript: A .DOC version of the revised manuscript
- Figures: Editable, high-resolution, individual figure files are required at revision, TIFF or EPS files are preferred

For complete guidelines on revision requirements, please see the journal Submission and Review Process requirements at

https://journals.asm.org/journal/Spectrum/submission-review-process. Submissions of a paper that does not conform to Microbiology Spectrum guidelines will delay acceptance of your manuscript. "

Please return the manuscript within 60 days; if you cannot complete the modification within this time period, please contact me. If you do not wish to modify the manuscript and prefer to submit it to another journal, please notify me of your decision immediately so that the manuscript may be formally withdrawn from consideration by Microbiology Spectrum.

If your manuscript is accepted for publication, you will be contacted separately about payment when the proofs are issued; please follow the instructions in that e-mail. Arrangements for payment must be made before your article is published. For a complete list of **Publication Fees**, including supplemental material costs, please visit our website.

Corresponding authors may join or renew ASM membership to obtain discounts on publication fees. Need to upgrade your membership level? Please contact Customer Service at Service@asmusa.org.

Thank you for submitting your paper to Microbiology Spectrum.

The authors developed a new method to rapidly detect hvKp in clinical samples based on recombinase-aided amplification, and thought this approach is a powerful tool to solve the challenge of timely detection of hvKp infections in clinical and surveillance settings. There are comments which the authors should clarify.

Major comments:

- 1. Considerable English revision is needed. There are too many chinglish expression in manuscript.
- 2. The title is focused on the rapid detection of hvKp. However, in manuscript, the authors also have written much information about phenotype and molecular characteristics of hvKp. These contents are not in accordance with the topic of this manuscript. Please clarify this.
- 3. There is on hvKp/cKp standard strains were used to evaluate the specificity in the study. It is not enough to use only plasmid containing genes *peg-344* and *rmpA*, at least two standard strains of hvKp and cKp should also be examined.
- 4. It is not convincing to evaluate the RAA assay in clinical detection of hvKp only using clinical isolates. I suggest the authors to check the ability of the developed assay from clinical samples, such as, blood, saliva and nasopharyngeal swabs.
- 5. The authors should analyze the economics of this assay and explain in discussion, since the authors thought the cost of RAA assay is lower than real-time PCR or other nucleic acid-based detection methods in discussion.

Minor comments:

- 1. Line 108: Delete the "peg-344 and rmpA"
- 2. Line 114: Delete the "using the RAA assay developed"
- 3. Line 114-116: Refine and rewrite this sentence.
- 4. Line 117: Change "The optimal reaction temperature range" to "The range of the reaction temperature".
 - 5. Line 121: Change "Fluorescence" to "Fluorescent".
 - 6. Line 123: Delete "evaluated"
 - 7. Line 126: Add "analysis" before the word "of"
- 8. Line 130: The results from conventional PCR and real-time PCR could not be reviewed in Figure 3. Suggest the authors add these figures in Figure 3.
- 9. Line 131: Here, the unit of the concentration of recombinant plasmid is "copies/reaction", which is not in line with "copies/µl" in Line 135. Please the authors clarify.
 - 10. Line 141-143: Refine and rewrite this sentence.
 - 11. Line 148: Delete the "assay" before the word "results".
 - 12. Line 149: Delete "which".
 - 13. Line 159-161: Refine and rewrite this sentence.
 - 14. Line 198-199: Refine and rewrite this sentence.
 - 15. Line 202-204: Refine and rewrite this sentence.

Dear editor,

Thank you for your e-mail informing us that our manuscript will benefit from being revised according to the suggestions of the reviewers.

We would like to thank all the reviewers for their very kind and constructive comments on our manuscript. Our responses to their comments are as follows:

Changes in the revised manuscript are red.

Reviewer comments:

Reviewer #1 (Comments for the Author):

Manuscript is sound in methodology and analysis. Appropriate controls were implemented.

Line 74 Missing space between "siderophore" and "for"

Answer: Thanks for your suggestion. As suggested, we add space between "siderophore" and "for" (Line 82 in manuscript R1).

Line 133 lists RAA sensitivity as 20 copies while line 239 lists 10 copies

Answer: Thanks for your suggestion. As suggested, we revised 10 copies to 20 copies/reaction (Line 237 in manuscript R1).

Reviewer #2 (Comments for the Author):

The authors developed a new method to rapidly detect hvKp in clinical samples based on recombinase-aided amplification, and thought this approach is a powerful tool to solve the challenge of timely detection of hvKp infections in clinical and surveillance settings. There are comments which the authors should clarify.

Major comments:

1. Considerable English revision is needed.

Answer: Thanks for your suggestion. As suggested, we revised English of our

manuscript. We thank Michelle Kahmeyer-Gabbe, PhD, from Liwen Bianji (Edanz) (www.liwenbianji.cn) for editing the English text of a draft of this manuscript. Professor Michelle Kahmeyer-Gabbe's research areas include Microbiology, Biochemistry and Cell Biology, Complementary and Alternative Medicine. We hope that this manuscript is now conducive to readers' reading and understanding. We also uploaded the EdanzEditing Certificate as supplementary materials.

2. The title is focused on the rapid detection of hvKp. However, in manuscript, the authors also have written much information about phenotype and molecular characteristics of hvKp. These contents are not in accordance with the topic of this manuscript. Please clarify this.

Answer: Thank you for your suggestion. In addition to establishing a rapid RAA detection method, the analysis of phenotype and molecular characteristics for hvKp is also an important part in our manuscript. It can not only help us to verify that this RAA method can detect hvKp with different characteristics (serotype or STs), but also help us further understand the molecular characteristics of hvKp in China. In order to better describe our research, we revised the title to "Recombinase-Aided Amplification Assay for Rapid Detection of Hypervirulent *Klebsiella pneumoniae* (hvKp) and Characterization of the hvKp Pathotype".

- 3. There is on hvKp/cKp standard strains were used to evaluate the specificity in the study. It is not enough to use only plasmid containing genes peg-344 and rmpA, at least two standard strains of hvKp and cKp should also be examined.
- Answer: Thanks for your suggestion. In this study, the *K. pneumoniae* strain ATCC BAA-2146 was used as reference strain for cKp. The *K. pneumoniae* strain LA.045 isolated from a liver abscess patient and proved to be hypervirulent using mouse lethality assay was used as reference strain for hvKp. As suggested, we have supplemented this information in the material method (Line285-288) and other corresponding parts of the manuscript.
- 4. It is not convincing to evaluate the RAA assay in clinical detection of hvKp only using clinical isolates. I suggest the authors to check the ability of the developed assay from clinical samples, such as, blood, saliva and nasopharyngeal

swabs.

Answer: Thanks for your suggestion. In the revised manuscript, we added an additional 50 *K. pneumoniae* negative clinical samples to evaluate our method. The samples include 10 fecal obtained from healthy individuals, 20 nasopharyngeal swabs, and 20 blood. For the *K. pneumoniae* positive clinical samples, we not only detected the isolates, we also tested DNA obtained from the original clinical samples. In the corresponding part of the manuscript, we have made modifications.

5. The authors should analyze the economics of this assay and explain in discussion, since the authors thought the cost of RAA assay is lower than real-time PCR or other nucleic acid-based detection methods in discussion.

Answer: Thanks for your suggestion. As there is no commercialized kit for hvKp detection at present, it is not rigorous to estimate the cost only by raw materials used in different methods. So, we deleted this sentence in discussion.

Minor comments:

1. Line 108: Delete the "peg-344 and rmpA"

Answer: Thanks for your suggestion. As suggested, we delete the "peg-344 and rmpA" (Line 111 in manuscript R1).

2. Line 114: Delete the "using the RAA assay developed"

Answer: As suggested, we delete the "using the RAA assay developed "(Line 117-118 in manuscript R1).

3. Line 114-116: Refine and rewrite this sentence.

Answer: As suggested, we refine and rewrite this sentence (Line 115-116 in manuscript R1).

4. Line 117: Change "The optimal reaction temperature range" to "The range of the reaction temperature".

Answer: As suggested, we change "The optimal reaction temperature range" to "The range of the reaction temperature" (Line 120 in manuscript R1).

5. Line 121: Change "Fluorescence" to "Fluorescent".

Answer: As suggested, we change "Fluorescence" to "Fluorescent" (Line 121 in

manuscript R1).

6. Line 123: Delete "evaluated"

Answer: As suggested, we delete "evaluated" (Line 124 in manuscript R1).

7. Line 126: Add "analysis" before the word "of"

Answer: As suggested, we add "analysis" before the word "of" (Line 126 in manuscript R1).

8. Line 130: The results from conventional PCR and real-time PCR could not be reviewed in Figure 3. Suggest the authors add these figures in Figure 3.

Answer: As suggested, we add the results from conventional PCR (Figure 3B and 3E) and real-time PCR (Figure 3C and 3F) in the revised Figure 3.

9. Line 131: Here, the unit of the concentration of recombinant plasmid is "copies/reaction", which is not in line with "copies/ μ l" in Line 135. Please the authors clarify.

Answer: As suggested, we revised "copies/reaction" to "copies/µl" in Line 128 (manuscript R1).

10. Line 141-143: Refine and rewrite this sentence.

Answer: As suggested, we refine and rewrite this sentence (Line 141-143 in manuscript R1).

11. Line 148: Delete the "assay" before the word "results".

Answer: As suggested, we delete the "assay" before the word "results" (Line 149 in manuscript R1).

12. Line 149: Delete "which".

Answer: As suggested, we delete "which" (Line 149-151 in manuscript R1).

13. Line 159-161: Refine and rewrite this sentence.

Answer: As suggested, we refine and rewrite this sentence (Line 159-162 in manuscript R1).

14. Line 198-199: Refine and rewrite this sentence.

Answer: As suggested, we refine and rewrite this sentence (Line 197-199 in manuscript R1).

15. Line 202-204: Refine and rewrite this sentence.

Answer: As suggested, we refine and rewrite this sentence (Line 201-204 in manuscript R1).

1st Revision - Editorial Decision February 17, 2023

February 13, 2023

Prof. Jing Yuan Capital Institute of Pediatrics Department of Bacteriology No. 2 yabao road, Chaoyang District Beijing, Beijing 100020 China

Re: Spectrum03984-22R1 (Recombinase-Aided Amplification Assay for Rapid Detection of Hypervirulent *Klebsiella pneumoniae* (hvKp) and Characterization of the hvKp Pathotype)

Dear Prof. Jing Yuan:

Your manuscript has been accepted, and I am forwarding it to the ASM Journals Department for publication. You will be notified when your proofs are ready to be viewed.

The ASM Journals program strives for constant improvement in our submission and publication process. Please tell us how we can improve your experience by taking this quick Author Survey.

Publication Fees: We have partnered with Copyright Clearance Center to collect author charges. You will soon receive a message from no-reply@copyright.com with further instructions. For questions related to paying charges through RightsLink, please contact Copyright Clearance Center by email at ASM_Support@copyright.com or toll free at +1.877.622.5543. Hours of operation: 24 hours per day, 7 days per week. Copyright Clearance Center makes every attempt to respond to all emails within 24 hours. For a complete list of **Publication Fees**, including supplemental material costs, please visit ourwebsite.

ASM policy requires that data be available to the public upon online posting of the article, so please verify all links to sequence records, if present, and make sure that each number retrieves the full record of the data. If a new accession number is not linked or a link is broken, provide production staff with the correct URL for the record. If the accession numbers for new data are not publicly accessible before the expected online posting of the article, publication of your article may be delayed; please contact the ASM production staff immediately with the expected release date.

Corresponding authors may join or renew ASM membership to obtain discounts on publication fees. Need to upgrade your membership level? Please contact Customer Service at Service@asmusa.org.

Thank you for submitting your paper to Spectrum.

Sincerely,

MARK PANDORI Editor, Microbiology Spectrum

Journals Department American Society for Microbiology 1752 N St., NW Washington, DC 20036 E-mail: spectrum@asmusa.org